NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

NICE guidelines

Equality impact assessment

Drug misuse prevention: targeted interventions

The impact on equality has been assessed during guidance development according to the principles of the NICE equality policy.

1.0 Scope: before consultation

- 1.1 Have any potential equality issues been identified during the development of the draft scope, before consultation, and, if so, what are they?
- 1. The scope focuses on populations who are most likely to start using drugs, who are already experimenting with drugs, or who use drugs occasionally. The guideline will replace PH4 on interventions to reduce substance misuse among vulnerable young people
- 2. The groups specifically covered are children, young people and adults who are most likely to start using drugs, who are already experimenting with drugs, or who use drugs occasionally and this includes:
 - People who frequently go to nightclubs or festivals
 - Men who have sex with men
 - Those who are more vulnerable like:
 - a) people with mental health problems
 - b) people involved in commercial sex work or who are being sexually exploited
 - c) children and young people who are not in education or training, including school excludes and truants
 - d) children and young people whose parents use drugs
 - e) children and young people who are looked after

3. It will not cover:

- People who already dependent on drugs or who use drugs regularly and excessively
- People who are in prison or young offender institutions
- People who are part of on-going drug treatment and recovery
- 4. A focus on those 'who are more vulnerable" potentially means that those who are less vulnerable may not be considered fully in the guidance. This potentially raises a number of equality issues in groups considered less vulnerable across the protected characteristics. The focus of the guidance is on settings and communities and thus does not set out to 'discriminate' on the basis of the protected characteristics and the scope is not suggesting that the risk of drug misuse is not present in these less vulnerable communities/populations. The scope has focused on those who are more vulnerable based on the evidence reported by other publications for example "Drug Misuse: Finding from the 2012 to 2013 Crime Survey for England and Wales (CSEW, 2014)" and "New psychoactive substances in England: a review of the evidence (Home Office, 2014)"
- 5. Excluding "People who are in prison or young offender institutions" can potentially raise a number of equality issues among people in prison settings. However, this guideline starts from the assumption that prisoners in the United Kingdom are already in a prevention/rehab scheme or system.
- 6. The key activities outlined in the scope include increasing awareness. Within the population categorised as more vulnerable there may be potential equality issues regarding the way in which this is done. Consideration needs to be given to the format of information delivery for example those with visual impairments or those for whom English is not their first language. This should be considered in the guideline development process.
- 7. There is also a potential equality issue regarding access to interventions for transient communities such as homeless people and Gypsy, Roma and Traveller communities. The scope and subsequent guideline development process should be aware of this and make provision to consider this in the guidance development process

1.2 What is the preliminary view on the extent to which these potential equality issues need addressing by the Committee? For example, if population groups, treatments or settings are excluded from the scope, are these exclusions justified – that is, are the reasons legitimate and the exclusion proportionate?

The committee will need to consider the above issues particularly in terms of the evidence base and thinking about settings for the delivery of interventions to reflect some protected characteristics.

The focus of the scope on who are more vulnerable in the population is justified. The exclusion of "people who already dependent on drugs or who use drugs regularly and excessively" and of "people who are part of on-going drug treatment and recovery" is justified by the fact that this guideline focusses on the prevention phase only and consequently it excludes people who already use drugs. The exclusion of prisoners is justified given the existing prevention programmes currently universally offered to people in prison in England.

Completed by Emanuele Mortara

Date - 2th February 2015

Approved by NICE

2.0 Scope: after consultation (To be completed by the developer and submitted with the final scope)

2.1 Have any potential equality issues been identified during consultation, and, if so, what are they?

Stakeholders identified several groups who they thought should be specifically mentioned within the scope, including

- Young people in contact with youth justice
- Roma
- Trans people
- BME groups

• People who are subject to or perpetrate domestic violence.

It was also felt that the exclusion of prisoners and people already in drug treatment in the draft scope was unreasonable.

2.2 Have any changes to the scope been made as a result of consultation to highlight potential equality issues?

As a result of stakeholder consultation, the exclusion of people already in structured treatment programmes has been removed. The lists of exemplars have not been changed to incorporate the other groups mentioned, partly due to a lack of reliable data, but also because the examples within the scope are clearly marked as examples. None of the groups mentioned are actually excluded, they are simply not specifically referred to in the examples. Prisons as a setting continue to be excluded since they will be covered in the NICE guideline 'Mental health of adults in contact with the criminal justice system'.

2.3 Is the primary focus of the guideline a population with a specific disabilityrelated communication need?

If so, is an alternative version of the 'Information for the Public' document recommended?

If so, which alternative version is recommended?

The alternative versions available are:

- large font or audio versions for a population with sight loss;
- British Sign Language videos for a population who are deaf from birth;
- 'Easy read' versions for people with learning disabilities or cognitive impairment.

The primary focus of the guideline is not a population with a specific disability-related communication need.

Date 22 April 2015

Approved by NICE quality assurance lead Simon Ellis

3.0 Guideline development: before consultation (to be completed by the developer before draft guideline consultation)

3.1 Have the potential equality issues identified during the scoping process been addressed by the Committee, and, if so, how?

The committee have addressed the potential equality issues identified during the scoping process.

People who are already dependent on drugs or who use drugs regularly and excessively are already covered by NICE's guidelines on treatment and care for people who misuse drugs. The committee acknowledged that it is difficult to determine whether some approaches are treatment or prevention.

The explicit exclusion of prisoners and people in institutions for young offenders was removed from the scope but the exclusion of custodial settings was maintained. Therefore studies of interventions delivered in prisons or institutions for young offenders were excluded from the evidence review. This is because NICE is currently developing a clinical guideline on the Mental health of adults in contact with the criminal justice system. In addition, there are existing prevention programmes currently universally offered to inmates in England.

People who are part of on-going drug treatment and recovery have been included in the development of this guideline. This allows for the fact that someone may be receiving treatment for one drug but can still receive interventions to prevent their misuse of another drug.

3.2 Have any **other** potential equality issues (in addition to those identified during the scoping process) been identified, and, if so, how has the Committee addressed them?

The committee discussed that the remit for this guideline only covered one aspect of drug misuse. In line with the scope, the evidence base on universal approaches or wider determinants of drug misuse was not considered. The committee noted the

3.2 Have any **other** potential equality issues (in addition to those identified during the scoping process) been identified, and, if so, how has the Committee addressed them?

importance of considering the recommendations in this guideline alongside others on the prevention and management of drug misuse (for more information see NICE
pathway on drug misuse). The committee discussed that wider determinants, such as housing, education and employment opportunities, social support and resilience can have a fundamental impact both on risk of drug misuse but also the effectiveness of interventions to prevent use or minimise harm. Therefore, the committee developed the recommendations in the guideline on the assumption that they would be considered alongside other relevant guidance produced by NICE such as guidelines on social and emotional wellbeing in primary and secondary education, alcohol use disorders, looked after children and young people and community engagement.

The committee discussed that many terms used in the literature to describe drug use are subjective and often used inconsistently or inter-changeably (for example, terms such as 'use', 'misuse', 'occasional' or 'recreational', 'dependency', and 'abuse'). They discussed that the definition of 'recreational' use in particular was subjective. For example, fortnightly use of cannabis as an adult might be considered recreational, but it may not be in a child or young person.

The committee were aware that some people who use drugs do not identify as drug users. This can include people who use image- or performance-enhancing drugs, or new psychoactive substances (also known as 'legal highs').

This guideline does not cover all of the vulnerable groups that were covered in previous NICE guidance PH4 on <u>Substance misuse interventions</u>. The current guideline has focused on those who are more vulnerable based on extensive scoping searches, crime statistics, stakeholder comments and an initial sift of the evidence.

The committee identified a further potential equity issue regarding people being in more than 1 at-risk group. The committee discussed this at length in the committee meetings. They highlighted that studies included in the evidence review for a particular at risk group may include people who fall into more than 1 at risk group, for example, 'children and young people whose parents use drugs' and 'looked after children and young people'. Text has been added to the evidence review to ensure that this is explicit. The discussion section of the guideline also has text to reflect this issue.

Studies of pregnant women were excluded from the evidence review for this

3.2 Have any **other** potential equality issues (in addition to those identified during the scoping process) been identified, and, if so, how has the Committee addressed them?

guideline. There is existing NICE guidance on <u>Pregnancy and complex social</u> <u>factors</u>, which includes substance misuse in pregnant women. General antenatal and postnatal care is covered in existing NICE guidance. In addition, it was agreed that studies on drug use in pregnant women are most likely to be studies of treatment for dependent drug users rather than prevention studies.

3.3 Were the Committee's considerations of equality issues described in the consultation document, and, if so, where?

The committee's considerations of equality issues highlighted in section 3.2 were described in 'the committee's discussion' section of the guideline document.

The issue around wider determinants was addressed under the 'approach of this guideline' heading, and also mentioned in the 'existing NICE guidance' section.

People who may not identify themselves as drug users are discussed in the discussion section for recommendations 1.5.1 and 1.5.2. There is also a research recommendation for further research into this group.

The differences in at-risk groups between the current guideline and the previous guideline (Substance misuse interventions for vulnerable under 25s [PH4]) are discussed in the 'update information' section.

The committee agreed not to use the term 'foster parents' in the recommendations as it implied that some recommendations were specific to foster parents and others were not. The term 'parents and carers' is used in the recommendations to indicate that recommendations are aimed at either biological or foster parents (or carers) where appropriate.

The potential for people to belong to more than 1 at-risk group is described in the 'background and terminology' section, 'limitations of the effectiveness and acceptability evidence' section, 'overview of the cost effectiveness' section of the discussion section and the discussion sections for recommendations 1.2.1, 1.2.2 and 1.2.3, recommendations 1.3.2 and 1.3.3, and recommendation 1.3.4, 1.3.5, 1.3.6 and 1.3.7.

Pregnant women were not discussed in the guideline, however, the exclusion of studies of pregnant women is explained in evidence review 1 and evidence review 2

3.3 Were the Committee's considerations of equality issues described in the consultation document, and, if so, where?

(see section 3.2 of the EIA).

3.4 Do the preliminary recommendations make it more difficult in practice for a specific group to access services compared with other groups? If so, what are the barriers to, or difficulties with, access for the specific group?

The preliminary recommendations do not make it more difficult in practice for a specific group to access services compared with other groups.

3.5 Is there potential for the preliminary recommendations to have an adverse impact on people with disabilities because of something that is a consequence of the disability?

The preliminary recommendations will not have an adverse impact on people with disabilities.

3.6 Are there any recommendations or explanations that the Committee could make to remove or alleviate barriers to, or difficulties with, access to services identified in questions 3.1, 3.2 or 3.3, or otherwise fulfil NICE's obligation to advance equality?

The committee has made specific recommendations for people who are not in contact with services, for example, because they do not identify as drug users (recommendations 1.5.1, 1.5.2 and 1.5.3). No other barriers to or difficulties with access to services were identified.

Completed by Developer: Ella Novakovic

Date: 30th June 2016

Approved by NICE quality assurance lead: Stephanie Fernley

Date: 5th July 2016

4.0 Final guideline (to be completed by the Developer before GE consideration of final guideline)

4.1 Have any additional potential equality issues been raised during the consultation, and, if so, how has the Committee addressed them?

Stakeholders raised several additional potential equality issues during consultation.

- Stakeholders queried which age ranges are applicable to the definition of children and young people. A definition of young people has been added to the guideline, as follows "People aged 10 to 18. It also includes people aged up to 25 who have special educational needs or a disability (consistent with the Children and Families Act 2014)."
- Stakeholders highlighted that children who run away from home are a group at particular risk of drug misuse. The discussion section in the guideline has been amended to emphasise that there is wide variation in vulnerability to drug misuse within each of the groups at risk. The committee noted that the variation in vulnerability within the groups at risk means that different people in each group at risk have different needs, despite being in the same group. The committee agreed that not all people in a group at risk will use drugs. It commented that drug use is more likely in some people within each group at risk than in others, such as those in multiple groups at risk, whose personal circumstances put them at increased risk, who may already be using drugs on an occasional basis, or who may already be misusing another substance, such as alcohol.
- Stakeholders highlighted that domestic abuse can be an important precursor of drug misuse. Although people experiencing domestic abuse was not identified as an at risk group for the guideline, there will be some overlap with people who are homeless or unstably housed. The discussion has been amended to emphasise that is wide variation in vulnerability to drug misuse within each of the groups at risk. The committee noted that the variation in vulnerability within the groups at risk means that different people in each group at risk have different needs, despite being in the same group. In addition, the discussion states that there may be additional groups at risk that have not been considered in the guideline, and that people could be included in more than 1 group at risk.
- One stakeholder was concerned about the use of the term 'health literacy' in the guideline, as it assumes that some people would be labelled as 'health illiterate'.
 The term 'health literacy' has been removed from the guideline and replaced with 'levels of understanding about health'.
- Several stakeholders suggested that referral to specialist services should only

4.1 Have any additional potential equality issues been raised during the consultation, and, if so, how has the Committee addressed them?

take place with consent from the individual involved, and that the need for consent was not clear in the draft recommendations. The recommendation to refer people who are misusing drugs to specialist services has been reworded. The new recommendation states that, if a person is already misusing drugs, NICE's existing guidance on psychosocial interventions and opioid detoxification for drug misuse in people aged 16 years and older, needle and syringe programs, and diagnosis and management of alcohol-use disorders should be looked at.

- Stakeholders highlighted the heterogeneity of people within the at risk groups, in particular people who are lesbian, gay, bisexual or transgender. They highlighted that men who have sex with men and are involved in chemsex are at particular risk of drug misuse. The discussion has been amended to emphasise that there is wide variation in vulnerability to drug misuse within each of the groups at risk. The committee noted that the variation in vulnerability within the groups at risk means that different people in each group at risk have different needs, despite being in the same group.
- Stakeholders were concerned that there are barriers to accessing physical services for people who are lesbian, gay, bisexual or transgender. Stakeholders suggested that relevant web-based solutions would be appropriate for this group. The recommendation on providing information in different formats, including webbased information, has been emphasised.
- Stakeholders highlighted that prescription and over the counter drugs, solvents and volatile substances need to be added to the definition of drug misuse. These substances have been added to the definition of 'drugs' in the 'terms used' section of the guideline.
- Stakeholders suggested that the NICE guidance on physical health in prisons would be relevant to the current guideline. A reference to the NICE guideline on physical health of people in prisons has been added to the discussion.
- Stakeholders emphasised that tools for assessing the risk of drug misuse should be standardised. They felt that this would ensure equity of access to services. The committee were not able to recommend a particular assessment tool for a number of reasons. The committee noted that most tools that are available focus on people who already use drugs rather than those at risk of misusing drugs. Comments from stakeholders during consultation on the draft guideline highlighted several assessment tools that stakeholders believed could be useful. However, the committee noted that these tools may not fit easily into routine and opportunistic appointments provided by statutory and other services. Some of the

4.1 Have any additional potential equality issues been raised during the consultation, and, if so, how has the Committee addressed them?

tools focused on alcohol use rather than drug use. The committee was aware that there is no single tool that is suitable for use in all groups at risk and for assessing all vulnerabilities. It was concerned that recommending a tool for a particular group at risk may result in the tool being used with other groups at risk, which may not be appropriate. The draft recommendations stated that the approach should be locally agreed. The recommendation has been updated to state that the approach should also be consistent.

- Stakeholders highlighted that offering online sources of support may not be appropriate for people in at risk groups that may not have access to the internet, such as people who are homeless. The version of the guideline that went out for public consultation included a recommendation about providing information in different formats, including printed information, as part of a recommendation on the content of the information. This recommendation has been separated into two recommendations to emphasise that information should be provided in different formats, including printed information (recommendations 1.5.2 and 1.5.3).
- 4.2 If the recommendations have changed after consultation, are there any recommendations that make it more difficult in practice for a specific group to access services compared with other groups? If so, what are the barriers to, or difficulties with, access for the specific group?

No changes have been made since consultation that would make the recommendations more difficult in practice for specific groups to access services compared with other groups.

4.3 If the recommendations have changed after consultation, is there potential for the recommendations to have an adverse impact on people with disabilities because of something that is a consequence of the disability?

No changes have been made since consultation that would have an adverse impact on people with disabilities because of something that is a consequence of the disability. 4.4 If the recommendations have changed after consultation, are there any recommendations or explanations that the Committee could make to remove or alleviate barriers to, or difficulties with, access to services identified in questions 4.2, 4.3 and 4.4, or otherwise fulfil NICE's obligations to advance equality?

No barriers to or difficulties with access have been identified in questions 4.2 or 4.3.

4.5 Have the Committee's considerations of equality issues been described in the final guideline document, and, if so, where?

The committee's considerations of equality issues have been described in several places throughout the final guideline document.

The 'approach to this guideline' section includes discussion of vulnerability within the groups at risk, that different people in each group at risk have different needs, and that people could be included in more than 1 group at risk. The section also states that there may be additional groups at risk that have not been considered in the guideline.

A definition of young people has been added to the guideline in the 'terms used in this guideline' section.

The importance of safeguarding is discussed in the committee's discussion on recommendation 1.2.5.

The need for different approaches for different groups of people (such as verbally, printed leaflets or web-based information) is discussed in the committee's discussion on recommendations 1.4.2, 1.5.1 and 1.5.2.

The guideline has referred to other NICE guidance in relevant areas (for example, physical health in prisons, child maltreatment, and domestic violence) in the 'Existing NICE guidance' section.

Updated by Developer: Ella Novakovic

Date: 13th December 2016

Approved by NICE quality assurance lead: Stephanie Fernley

Date: 13th December 2016

5.0 After Guidance Executive amendments – if applicable (To be completed by appropriate NICE staff member after Guidance Executive)

5.1 Outline amendments agreed by Guidance Executive below, if applicable:
None

Approved by Developer: Ella Novakovic

Date: 21st February 2017

Approved by NICE quality assurance lead: Stephanie Fernley

Date: 21st February 2017